Author’s response to reviews

Title: Pulmonary embolism causing atrial fibrillation with slow ventricular response: a case report

Authors:

Mohammad Abdulsalam (msalam87@hotmail.com)

Mohammed Elganainy (mohammednabil_eg@hotmail.com)

Ahmad Abdulsalam (dr.ahmad.j.abdulsalam@gmail.com)

Version: 1 Date: 23 Nov 2018

Author's response to reviews:

Dear Editor,

As a response to your comments, the following corrections have been noted:

- the detailed echocardiographic data has been added to the text highlighting documenting high pulmonary pressures as well as massive dilatation of the right cardiac chambers (as is apparent on the chest x-ray).

-The text has been reviewed by a fluent English language reader and several changes have been made.

We thank your comments, and we await a positive reply.